Author’s response to reviews

Title: The patient voice: A survey of worries and anxieties during health system transition in HIV services in Vietnam

Authors:

Shoko Matsumoto (smatsumo@acc.ncgm.go.jp)
Hoai Dung Nguyen (dnguyenhai2024@gmail.com)
Dung Nguyen (dung.niitd@gmail.com)
Giang Tran (giangminh08@gmail.com)
Junko Tanuma (jtanuma@acc.ncgm.go.jp)
Daisuke Mizushima (dmizushi@acc.ncgm.go.jp)
Kinh Van Nguyen (kinhnv@nhtd.vn)
Shinichi Oka (oka@acc.ncgm.go.jp)

Version: 3 Date: 23 Dec 2019

Author’s response to reviews:

Assistant Editor Comments:

1. Ethics approval
In the Ethics approval and consent to participate statement of the Declarations, please provide the full name of the local institutional review board – i.e. please expand the abbreviation NHTD.

Thank you for your comment. The full name of the local institutional review board is “the Bio-medical Research Ethics Committees of National Hospital for Tropical Diseases.” According to your advice, we described without abbreviation in the manuscript.

In Ethics approval and consent to participate in Declarations:
The study was approved both by the Human Research Ethics Committee of the National Center for Global Health and Medicine (reference: NCGM-G-002537-01) and by the Bio-medical Research Ethics Committees of National Hospital for Tropical Diseases (reference:08/HDDD-NDTU).

*At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strike-throughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

I upload a clean version of manuscript with a separate file of Figures.